Author's response to reviews

Title: 5-allyl-7-gen-difluoromethoxychrys in enhances TRAIL-induced apoptosis in human lung carcinoma A549 cells

Authors:

Zhaohui Xie (xzh413@126.com)
Meifang Quan (quanmeifang@yeah.net)
Fei Liu (liufei5460@yahoo.com.cn)
Jianguo Cao (caojianguo2005@yahoo.com.cn)
Jiansong Zhang (jwc_zjs@126.com)

Version: 4 Date: 27 June 2011

Author's response to reviews: see over
Dear Editor:

Thanks for your great contributions to this article! In terms of your suggestions, we have made the last proofread about this article, the followings are some corrections about spelling mistakes and reference formats:

1. Page 3 line 2: “upregulating” to “up-regulating”;
2. Page 5 line 3: “considered” to “considered as”;
3. Page 5 last line: “line” to “lines”;
4. Page 10 line 4: “actions and mechanisms” to “effect and mechanism”;
5. Page 10 bottom line 5: “ligand receptor-dependent” to “ligand/receptor-dependent”;
6. Page 12 last line: “attenuated blocked” to “attenuated or blocked”;
7. Page 13 line 8: deleting “by AFMC treatment using human NSCLC lines”;
8. Page 16 line 1: “a half of all” to “half of all”;
9. Page 16 line 2: “against” to “to”; “Because” to “Although”;
10. Page 16 bottom line 1: “gavage of” to “gavage to”;
11. Recheck the reference part and list all authors of reference [6], [7], [10], [31];
12. Revise some small problems found in this article.

We are looking forward to your rechecking and precious suggestions. Thank you!